Author’s response to reviews

Title: Prognostic significance of anti-p53 and anti-KRas circulating antibodies in esophageal cancer patients treated with chemoradiotherapy

Authors:

Pierre Blanchard (pierre.blanchard@igr.fr)
Laurent Quero (laurent.quero@sls.aphp.fr)
Marie-Hélène Schlageter (marie-helene.schlageter@sls.aphp.fr)
Vincent Pacault (pacaultv@yahoo.fr)
Valérie Baruch-Hennequin (valerie.hennequin@sls.aphp.fr)
Christophe Hennequin (christophe.hennequin@sls.aphp.fr)

Version: 4  Date: 12 March 2012

Author’s response to reviews: see over
Dear Editor,

Please find the final version of our manuscript on “Prognostic significance of anti-p53 and anti-KRas circulating antibodies in esophageal cancer patients treated with chemoradiotherapy”.

To answer your last point, the use of the database had been approved by the institution’s ethics committee, but we had forgotten to mention it in the manuscript. We therefore added in the Methods section (paragraph Patients), as requested, a sentence stating: “The use of the patients’ clinical data was approved by the local research and ethics committee”.

Nothing else has been modified on the manuscript compared to the previous version.

Sincerely,

Pierre Blanchard.